Author's response to reviews

Title: Patterns in sedentary and exercise behavior and associations with overweight in 9-14-year-old boys and girls - a cross-sectional study-

Authors:

  Saskia J te Velde (s.tevelde@erasmusmc.nl)
  Ilse de Bourdeaudhuij (ilse.debourdeaudhuij@ugent.be)
  Inga Thorsdottir (ingathor@landspitali.is)
  Mette Rasmussen (M.Rasmussen@pubhealth.ku.dk)
  Maria Hagstromer (mahag@biosci.ki.se)
  Knut-Inge Klepp (k.i.klepp@medisin.uio.no)
  Johannes Brug (j.brug@erasmusmc.nl)

Version: 6 Date: 29 January 2007

Author's response to reviews:

Dear editor,

changes were made according the suggestions in your email dd 26 January 2007:
- running title is removed
- keywords are removed
- webaddress is removed from the text
- REFERENCES has been changed to References
- TABLES has been changed to Tables

kind regards
Saskia te Velde